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Protons on Target (since 2005) 

NuMI-MINOS Status Report – Nick Graf !

All Exp Mtg 
August 16, 2010!

1.07 x 1021 POT!
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NuMI-MINOS Status Report – Nick Graf !

Status of Shutdown Activities 

Last Week: Target NT-03 removed.  Underground water cooling 
upgrade completed.  Water cooling loop has been filled and is 
back on.  Building cooling and dehumidification should be back 
this afternoon. 

This Week: Install target NT-04 into the chase, complete beam 
scans, and prepare target hall for beam operations.  

All Exp Mtg 
August 16, 2010!
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MINOS Activities 

Control Room:  All electrical work finished.  All computers 
except OM upgraded to SLF 5.5.  Still some issues with Acnet. 

FD: Planned network upgrade postponed. 

ND:  Gateway computer replaced.  All filters for electronics racks 
replaced or cleaned.  Plan to power on detector and DAQ 
Tuesday.  
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All Exp Mtg 
June 14, 2010!

Beam Startup 

Originally expected to complete target scans no later than 
Tuesday with beam returning on Wednesday.  Scans delayed 
until Thursday/Friday due to problem with vacuum response of 
Lambertson, will need to be baked. 

Plan to start running shifts on Wednesday with night optional 
until beam returns next weekend.  


